Author’s response to reviews

Title: Risks and complications of transurethral resection of bladder tumors in patients receiving antiplatelet and/or anticoagulant therapy: a retrospective cohort study

Authors:

Tsuzumi Konishi (tsuzumi0203@gmail.com)
Satoshi Washino (suwajiisan@jichi.ac.jp)
Yuhki Nakamura (yuhki.dreamskyward@gmail.com)
Masashi Ohshima (m_ohshima_amihsho_m@yahoo.co.jp)
Kimitoshi Saito (lespaul991200@gmail.com)
Yoshiaki Arai (aria@nishiohmiya-hp.or.jp)
Tomoaki Miyagawa (sh2-miya@jichi.ac.jp)

Version: 4 Date: 29 Nov 2017

Author’s response to reviews:

Dear Editor,

We would like to thank the Editor for their helpful comments and for giving us the opportunity to make additional revisions to our manuscript. We have addressed these comments in the latest revised version, and attach our point-by-point responses below.

1) Methods:

We have moved the subsection ‘Anesthesia’ in the Methods section to results section (Page 9, line 8–13).

2) Declarations section:

According to your advice, we have changed our manuscript in the Declarations section (Page 16, line 5–9).

3) Copy edit:

We changed our manuscript in the discussion section (Page 15, line 6-7)